Author's response to reviews

Title: Promoting the uptake of evidence in challenging circumstances: practitioner experiences of implementing a co-created Injury Prevention Briefing to improve fire safety in the home. A qualitative study nested within a multi-centre randomised controlled trial.

Authors:

Kate Beckett Ms (Kate2.Beckett@uwe.ac.uk)
Trudy Goodenough Dr (Trudy.Goodenough@uwe.ac.uk)
Toity Deave Dr (Toity.Deave@uwe.ac.uk)
Sally Jaeckle (Sally.Jaeckle@bristol.gov.uk)
Lisa McDaid Ms (Lisa.Mcdaid@nnuh.nhs.uk)
Penny Benford Dr (Penny.Benford@nottingham.ac.uk)
Mike Hayes Mr (Mike.Hayes@capt.org.uk)
Elizabeth Towner Professor (Elizabeth.Towner@uwe.ac.uk)
Denise Kendrick Professor (Denise.Kendrick@Nottingham.ac.uk)

Version: 2 Date: 7 July 2014

Author's response to reviews: see over
To: The Editor
Implementation Science

Dear Editor

Re: Article submission MS: 3676529151347669 ‘Promoting the uptake of evidence in challenging circumstances: practitioner experiences of implementing a co-created Injury Prevention Briefing to improve fire safety in the home. A qualitative study nested within a multi-centre randomised controlled trial’.

Authors: Kate Beckett, Trudy Goodenough, Toity Deave, Sally Jaeckle, Lisa McDaid, Penny Benford, Mike Hayes, Elizabeth Towner, Denise Kendrick and the Keeping Children Safe Programme study group

Thank you for your e-mail requesting revisions to this article. We have made the changes requested including:

1. All tables have been placed at the end of the manuscript after the reference list
2. Figure 1 has been uploaded as a separate file, the title and legend have been included after the reference list.

Many thanks

Kate Beckett